Author's response to reviews

Title: A novel SNP analysis method to detect copy number alterations with an unbiased reference signal directly from tumor samples

Authors:

Alex Lisovich (all67@pitt.edu)
Uma R. Chandran (chandranur@upmc.edu)
Maureen A. Lyons-Weiler (lyonsweilerma@upmc.edu)
William A. LaFramboise (laframboisewa@upmc.edu)
Ashley R. Brown (arg39@pitt.edu)
Regina I. Jakacki (regina.jakacki@chp.edu)
Ian F. Pollack (ian.pollack@chp.edu)
Robert W. Sobol (rws9@pitt.edu)

Version: 4 Date: 25 January 2011

Author's response to reviews:

In response to your e-mail of January 21, 2011, we have reformatted the paper as requested and have made all requested changes.